Author’s response to reviews

Title: PRESERVATION OF RENAL FUNCTION IN CARDIAC SURGERY PATIENTS WITH LOW CARDIAC OUTPUT SYNDROME: LEVOSIMENDAN VS BETA AGONISTS

Authors:

JOSE LUIS GUERRERO ORRIACH (guerreroorriach@gmail.com)
ISABEL Arce (isanavarroarce@gmail.com)
PAULA RODRIGUEZ (paulahernandez81@gmail.com)
aida Ponferrada (aidaraigonp@gmail.com)
ALFREDO Manso (alfredomalomanso@gmail.com)
MARTA ALIAGA (martaram85@gmail.com)
ALICIA FERNANDEZ (aliramfdez@gmail.com)
JUAN JOSE BELMONTE (jjescalonabelmonte@gmail.com)
INMACULADA ESTEVEZ (inmabellido@gmail.com)
AURELIO LUQUE (jagomez@uma.es)
RAFAEL SERRANO (rafbaser@gmail.com)
CARMEN MEDINA (carmentm_86@hotmail.com)
MANUEL navarro (mrubionav@gmail.com)
jose mañas (jose.cruz.sspa@juntadeandalucia.es)

Version: 4 Date: 25 Jun 2019

Author’s response to reviews:

Good news for us that reviewers have agree the suitable of our article in your journal.

About the 2 questions of editor, first one we have tried to explain in the text and in the previous mail that we send him, 2 days before; is enough our explication about the way that we did the methods (could you need something more)?, we will explain all the questions that he needs to try to clarify the doubts about this part of our study.
But about the register we are afraid that all the registers are about clinical trials, quasiexperimental studies dont have international registers ony clinical trials, this study isnt a clinical trial, but has been approved by Ethical C. and has a local and national registration.